Author's response to reviews

Title: Improved Technique of Mitomycin C Application May Decrease the Encapsulated Cyst Incidence Following Ahmed Glaucoma Valve Implantation in Refractory Glaucoma Patients

Authors:

Minwen Zhou (zmw8008@163.com)
Wei Wang (will.wong@qq.com)
Wenbin Huang (cyhwenb@gmail.com)
Xiulan Zhang (zhangxl2@mail.sysu.edu.cn)

Version: 2 Date: 31 December 2013

Author's response to reviews:

1. When you received ethical approval for this study and what they were specifically approving.
   Answer# we had added the time in the manuscript (page 3).

2. How did you decide which patients received the difference surgical treatments?
   Answer# we had added this content in the manuscript (page 4).